\begin{table}%t2
\caption{Compilation of D/H, D/O, titanium and iron data. Errors bars 
are quoted in the usual way with negative errors as indices and 
positive errors as exponants. References of TiII~data used in the 
correlations are: (a)~Prochaska et~al. (\cite{Proc05}), (b)~Ellison et~al. (\cite{Elli07}), (c)~this work. Targets in common with Ellison et~al. (\cite{Elli07}) are marked with an asterix. The column-densities derived by these authors are log~(TiII)~= 11.10~$\pm$ 0.02; 11.31~$\pm$ 0.01; 11.12~$\pm$ 0.02; 
11.05~$\pm$ 0.04; 11.07~$\pm$ 0.04 for $\gamma^{2}$~Vel; $\theta$~Car; 
$\alpha$~Cru; $\beta$~Cen and $\zeta$~Pup respectively and are in 
agreement with our determinations within uncertainties.
D, H, D/H, D(Fe) are the values compiled by Linsky et~al. (\cite{Lins06}), 
while D, O, D/O are those compiled by Oliveira et~al. (\cite{Olive06}). The 
three targets with a very high~D/O and discussed in Sect.~3.2 are 
marked with a~\#; the first D/O~value is the published value used in 
Fig.~\ref{fig9}a. The second number is the reduced value used in Fig.~\ref{fig9}a (see Sect.~3.2).}\label{table2}
\par
%\centering
\small
\begin{tabular}{c c c c l l l l c }      
\hline\hline
%&&&&&&&& \\[-8pt]
Target & log(NHI) & log~(DI) & D/H(ppm) & DI/OI $\times$ 10$^{2}$ & D(Fe) & 
log~(TiII)  & TiII/HI $\times$ 10$^{10}$  & Ref.\\
\hline
%&&&&&&&& \\[-8pt]
SiriusB & $\rm 17.6^{ 0.14 }_{ 0.12 }$& $\rm 12.81^{ 0.09 }_{ 0.09 }$& $\rm 16.2^{ 7.2 }_{ 6.4 }$& $\rm 3.9^{ 0.8 }_{ 0.8 }$& $\rm -1.11^{ 0.12 }_{ 0.14 
}$&      &  $\rm ^{   }_{   }$& \\
36Oph & $\rm 17.85^{ 0.075 }_{ 0.075 }$& $\rm 13.025^{ 0.01 }_{ 0.01 }$& $\rm 15^{ 2.5 }_{ 2.5 }$&   & $\rm -0.65^{ 0.27 }_{ 0.27 }$& 
&   &  \\
$\epsilon$ Eri & $\rm 17.875^{ 0.035 }_{ 0.035 }$& $\rm 13.03^{ 0.03 }_{ 
0.03 }$& $\rm 14.3^{ 1.6 }_{ 1.6 }$&   & $\rm -1.07^{ 0.1 }_{ 
0.1 }$&      &   &  \\
31 Com & $\rm 17.884^{ 0.03 }_{ 0.03 }$& $\rm 13.19^{ 0.025 }_{ 0.025 }$& 
$\rm 20.2^{ 1.9 }_{ 1.9 }$&   & $\rm -0.78^{ 0.1 }_{ 0.1 }$& 
&   &  \\
HZ43 & $\rm 17.93^{ 0.03 }_{ 0.03 }$& $\rm 13.15^{ 0.02 }_{ 0.02 }$& 16.6 
$^{ 1.4 }_{ 1.4 }$& $\rm 4.6^{ 0.5 }_{ 0.5 }$& $\rm -1.21^{ 0.04 }_{ 0.04 
}$&      &   &  \\
Procyon & $\rm 18.06^{ 0.05 }_{ 0.05 }$& $\rm 13.26^{ 0.027 }_{ 0.027 }$& 
$\rm 16^{ 2 }_{ 2 }$&   & $\rm -1.24^{ 0.05 }_{ 0.05 }$& 
&   &  \\
$\beta$ Cas & $\rm 18.132^{ 0.025 }_{ 0.025 }$& $\rm 13.36^{ 0.03 }_{ 0.03 
}$& $\rm 16.9^{ 1.6 }_{ 1.6 }$&   & $\rm -1.22^{ 0.1 }_{ 0.1 
}$&      &   &  \\
G191-B2B & $\rm 18.18^{ 0.09 }_{ 0.09 }$& $\rm 13.4^{ 0.035 }_{ 0.035 }$& 
$\rm 16.6^{ 4.1 }_{ 4.1 }$& $\rm 3.5^{ 0.4 }_{ 0.4 }$& $\rm -0.58^{ 0.09 }_{ 
0.09 }$&      &   &  \\
$\beta$ Gem & $\rm 18.261^{ 0.037 }_{ 0.037 }$& $\rm 13.43^{ 0.05 }_{ 0.05 
}$& $\rm 14.8^{ 2.2 }_{ 2.2 }$&   & $\rm -1.29^{ 0.05 }_{ 
0.05 }$&      &   &  \\
Capella & $\rm 18.239^{ 0.035 }_{ 0.035 }$& $\rm 13.44^{ 0.02 }_{ 0.02 }$& 
$\rm 15.9^{ 1.5 }_{ 1.5 }$& $\rm 2.6^{ 1.2 }_{ 1.2 }$& $\rm -1.2^{ 0.04 }_{ 
0.04 }$&      &   &  \\
$\alpha$ Tri & $\rm 18.327^{ 0.035 }_{ 0.035 }$& $\rm 13.45^{ 0.05 }_{ 0.05 
}$& $\rm 13.3^{ 2 }_{ 2 }$&   & $\rm -1.13^{ 0.2 }_{ 0.2 }$& 
&   &  \\
WD0621 376 & $\rm 18.7^{ 0.15 }_{ 0.15 }$& $\rm 13.85^{ 0.045 }_{ 0.045 }$& 
$\rm 14.1^{ 6 }_{ 6 }$& $\rm 3.9^{ 0.6 }_{ 0.6 }$& $\rm -1.21^{ 0.2 }_{ 0.2 
}$&     &   &  \\
WD2211 495 & $\rm 18.76^{ 0.15 }_{ 0.15 }$& $\rm 13.94^{ 0.05 }_{ 0.05 }$& 
$\rm 15.1^{ 6.5 }_{ 6.5 }$& $\rm 4^{ 0.6 }_{ 0.6 }$& $\rm -0.96^{ 0.16 }_{ 
0.16 }$&      &   &  \\
WD1634 573 & $\rm 18.85^{ 0.06 }_{ 0.06 }$& $\rm 14.05^{ 0.025 }_{ 0.025 
}$& $\rm 15.8^{ 2.5 }_{ 2.5 }$& $\rm 3.5^{ 0.3 }_{ 0.3 }$& $\rm -1.22^{ 0.09 
}_{ 0.09 }$&      &   &  \\
$\alpha$ Vir & $\rm 19^{ 0.1 }_{ 0.1 }$& $\rm 14.2^{ 0.2 }_{ 0.1 }$& 15.8 
$^{ 10.8 }_{ 5.8 }$& $\rm 4.2^{ 2 }_{ 2 }$& $\rm -1.29^{ 0.1 }_{ 0.12 }$& 
10.49 $\pm$ 0.13 & $\rm 32.4^{ 14.4 }_{ 10 }$& c \\
GD246 & $\rm 19.11^{ 0.025 }_{ 0.025 }$& $\rm 14.29^{ 0.045 }_{ 0.045 }$& 
$\rm 15.1^{ 1.9 }_{ 1.9 }$& $\rm 4.2^{ 0.6 }_{ 0.6 }$& $\rm -1.26^{ 0.06 }_{ 
0.06 }$&      &   &  \\
$\lambda$ Sco & $\rm 19.23^{ 0.03 }_{ 0.03 }$& $\rm 14.11^{ 0.09 }_{ 0.07 
}$& $\rm 7.6^{ 1.4 }_{ 1.8 }$& $\rm 1.8^{ 0.4 }_{ 0.3 }$& $\rm -1.64^{ 0.05 
}_{ 0.05 }$&      &   &  \\
$\beta$ Cen & $\rm 19.63^{ 0.1 }_{ 0.1 }$& $\rm 14.7^{ 0.2 }_{ 0.2 }$& $\rm 11.7^{ 7.5 }_{ 7.5 }$&   & $\rm -1.16^{ 0.11 }_{ 0.11 }$& 
10.99 $\pm$ 0.04 & $\rm 24^{ 6.9 }_{ 5.4 }$& *c \\
$\gamma^{2}$ Vel & $\rm 19.71^{ 0.026 }_{ 0.026 }$& $\rm 15.05^{ 0.03 }_{ 
0.03 }$& $\rm 21.8^{ 2.1 }_{ 2.1 }$&   & $\rm -1.21^{ 0.11 
}_{ 0.11 }$& 11.11 $\pm$ 0.035 & $\rm 26.3^{ 2.5 }_{ 2.3 }$&*c \\
BD284211 & $\rm 19.85^{ 0.02 }_{ 0.02 }$& $\rm 14.99^{ 0.025 }_{ 0.025 }$& 
$\rm 13.9^{ 1 }_{ 1 }$& $\rm 4.7^{ 0.4 }_{ 0.4 }$& $\rm -1.2^{ 0.1 }_{ 0.1 }$& 
11.08 $\pm$ 0.08 & $\rm 17^{ 3.4 }_{ 2.9 }$& a \\
$\alpha$ Cru & $\rm 19.85^{ 0.07 }_{ 0.1 }$& $\rm 14.95^{ 0.05 }_{ 0.05 }$& 
$\rm 12.6^{ 3.6 }_{ 2.7 }$&   & $\rm -1.3^{ 0.14 }_{ 0.12 }$& 
11.12 $\pm$ 0.035 & $\rm 19.5^{ 3.9 }_{ 3.3 }$&*c \\
$\mu$ Col & $\rm 19.86^{ 0.015 }_{ 0.015 }$& $\rm 14.7^{ 0.3 }_{ 0.1 }$& 
$\rm 6.9^{ 6.9 }_{ 1.8 }$&   & $\rm -1.18^{ 0.02 }_{ 0.02 }$& 
11.50 $\pm$ 0.02 & $\rm 45.7^{ 3.3 }_{ 3.1 }$& c \\
Lan 23 & $\rm 19.89^{ 0.25 }_{ 0.04 }$& $\rm 15.23^{ 0.065 }_{ 0.065 }$& 
$\rm 21.9^{ 17.4 }_{ 4.1 }$& $\rm 3.2^{ 1.6 }_{ 1.6 }$& $\rm -1.31^{ 0.07 }_{ 
0.25 }$&      &   &  \\
$\zeta$ Pup & $\rm 19.96^{ 0.026 }_{ 0.026 }$& $\rm 15.11^{ 0.06 }_{ 0.06 
}$& $\rm 14^{ 2.3 }_{ 2.3 }$&   & $\rm -1.28^{ 0.17 }_{ 0.07 
}$& 11.16 $\pm$ 0.06 & $\rm 16.6^{ 2.9 }_{ 2.5 }$& *c \\
TD1 32709 & $\rm 20.03^{ 0.1 }_{ 0.1 }$& $\rm 15.3^{ 0.05 }_{ 0.05 }$& $\rm 18.6 ^{ 5.3 }_{ 5.3 }$&7.59;$\rm3.75^{ 2.17 }_{ 1.76 }$& $\rm -1.53^{ 0.14 }_{ 
0.14 }$&      &   &\#  \\
WD1034+001 & $\rm 20.07^{ 0.07 }_{ 0.07 }$& $\rm 15.4^{ 0.07 }_{ 0.07 }$& 
$\rm 21.4^{ 5.3 }_{ 4.5 }$& $\rm 6.31;3.15^{ 1.79 }_{ 1.38 }$& $\rm -1.42^{ 
0.12 }_{ 0.12 }$& 11.1 $\pm$ 0.2 & $\rm 10.7^{ 6.7 }_{ 4.1 }$& b\# \\
BD+39 3226 & $\rm 20.08^{ 0.09 }_{ 0.09 }$& $\rm 15.15^{ 0.05 }_{ 0.05 }$& 
$\rm 11.7^{ 3.1 }_{ 3.1 }$& $\rm 5.62;2.81^{ 1.61 }_{ 1.31 }$& $\rm -1.38^{ 
0.11 }_{ 0.11 }$& 11.49 $\pm$ 0.08 &   & b\# \\
Feige110 & $\rm 20.14^{ 0.065 }_{ 0.1 }$& $\rm 15.47^{ 0.03 }_{ 0.03 }$& 
$\rm 21.38^{ 5.7 }_{ 3.8 }$& $\rm 2.6^{ 0.5 }_{ 0.5 }$&  & 
11.59 $\pm$ 0.03 & $\rm 28.2^{ 4.9 }_{ 4.2 }$& a \\
$\iota$ Ori & $\rm 20.16^{ 0.06 }_{ 0.07 }$& $\rm 15.3^{ 0.04 }_{ 0.05 }$& 
$\rm 14.13^{ 2.8 }_{ 2.8 }$& $\rm 3.5^{ 0.8 }_{ 0.8 }$& $\rm -1.41^{ 0.21 }_{ 
0.16 }$& 11.31 $\pm$ 0.03 & $\rm 14.1^{ 2.5 }_{ 2.1 }$& a \\
$\gamma$ Cas & $\rm 20.16^{ 0.08 }_{ 0.1 }$& $\rm 15.15^{ 0.04 }_{ 0.05 }$& 
$\rm 9.8^{ 2.3 }_{ 2.7 }$& $\rm 2.5^{ 0.4 }_{ 0.4 }$& $\rm -1.53^{ 0.14 }_{ 
0.18 }$&      &   &  \\
$\delta$ Ori & $\rm 20.19^{ 0.03 }_{ 0.03 }$& $\rm 15.06^{ 0.06 }_{ 0.04 
}$& $\rm 7.4^{ 1.2 }_{ 0.9 }$& $\rm 2.5^{ 0.4 }_{ 0.4 }$& $\rm -1.56^{ 0.04 
}_{ 0.04 }$& 11.15 $\pm$ 0.04 & $\rm 9.1^{ 1.1 }_{ 1 }$& a \\
$\theta$ Car & $\rm 20.28^{ 0.1 }_{ 0.1 }$& $\rm 14.98^{ 0.18 }_{ 0.21 }$& 
$\rm 5^{ 2.9 }_{ 3.4 }$&   & $\rm -1.54^{ 0.11 }_{ 0.11 }$& 
11.31 $\pm$ 0.023 & $\rm 11.2^{ 2.9 }_{ 2.3 }$& *c \\
$\epsilon$ Ori & $\rm 20.4^{ 0.08 }_{ 0.1 }$& $\rm 15.25^{ 0.05 }_{ 0.05 
}$& $\rm 6.3^{ 1.8 }_{ 1.5 }$& $\rm 1.9^{ 0.3 }_{ 0.3 }$& $\rm -1.7^{ 0.14 }_{ 
0.12 }$& 11.4 $\pm$ 0.03 & $\rm 10.0^{ 2.3 }_{ 1.9 }$& a \\
PG 0038+199 & $\rm 20.48^{ 0.04 }_{ 0.04 }$& $\rm 15.76^{ 0.04 }_{ 0.04 }$& 
$\rm 19.1^{ 2.6 }_{ 2.6 }$& $\rm 2.4^{ 1 }_{ 0.5 }$& $\rm -1.51^{ 0.05 }_{ 
0.04 }$&      &   &  \\
JL9 & $\rm 20.78^{ 0.05 }_{ 0.05 }$& $\rm 15.78^{ 0.06 }_{ 0.06 }$& $\rm 10^{ 
1.9 }_{ 1.9 }$& $\rm 1.9^{ 0.8 }_{ 0.8 }$& $\rm -1.54^{ 0.1 }_{ 0.1 }$& 
11.88 $\pm$ 0.1 & $\rm 13.2^{ 3.8 }_{ 2.9 }$& c \\
HD 195965 & $\rm 20.95^{ 0.025 }_{ 0.025 }$& $\rm 15.88^{ 0.07 }_{ 0.07 }$& 
$\rm 8.51^{ 1.6 }_{ 1.6 }$& $\rm 1.3^{ 0.3 }_{ 0.3 }$& $\rm -1.59^{ 0.03 }_{ 
0.03 }$& 12.02 $\pm$ 0.02 & $\rm 11.7^{ 0.8 }_{ 0.8 }$& a \\
LS1274 & $\rm 20.98^{ 0.04 }_{ 0.04 }$& $\rm 15.86^{ 0.09 }_{ 0.09 }$& 7.6 
$^{ 1.9 }_{ 1.9 }$& $\rm 1.8^{ 0.5 }_{ 0.5 }$& $\rm -1.62^{ 0.08 }_{ 0.08 
}$& 11.68 $\pm$ 0.04 & $\rm 5.2^{ 0.8 }_{ 0.7 }$& c \\
HD 191877 & $\rm 21.05^{ 0.05 }_{ 0.05 }$& $\rm 15.94^{ 0.11 }_{ 0.06 }$& 
$\rm 7.8^{ 2.4 }_{ 1.5 }$& $\rm 2.5^{ 1 }_{ 1 }$& $\rm -1.55^{ 0.05 }_{ 0.05 
}$& 12.24 $\pm$ 0.02 & $\rm 15.5^{ 1.9 }_{ 1.7 }$& a \\
HD 41161 & $\rm 21.08^{ 0.09 }_{ 0.09 }$& $\rm 16.41^{ 0.05 }_{ 0.05 }$& 
$\rm 21.4^{ 5.7 }_{ 4.5 }$&   &  & 12.28 $\pm$ 0.04 & $\rm 15.8^{ 4.1 }_{ 3.3 }$& b \\
HD 53975 & $\rm 21.14^{ 0.06 }_{ 0.06 }$& $\rm 16.15^{ 0.07 }_{ 0.07 }$& 
$\rm 10.2^{ 2.4 }_{ 2 }$&   &  & 12.13 $\pm$ 
0.04 & $\rm 9.8^{ 1.7 }_{ 1.5 }$& b \\
HD 90087 & $\rm 21.22^{ 0.05 }_{ 0.05 }$& $\rm 16.16^{ 0.06 }_{ 0.06 }$& 
$\rm 8.7^{ 1.7 }_{ 1.7 }$& $\rm 1.7^{ 0.4 }_{ 0.4 }$& $\rm -1.45^{ 0.05 }_{ 
0.05 }$&      &   &  \\
\hline
\end{tabular}
\end{table}